Author's response to reviews

Title: Web-based Family History Assessment Tools for Colorectal Cancer (CRC) Risk Analysis - Development and Comparison

Authors:

Michael Schultz (michael.schultz@otago.ac.nz)
Steven Bohwan Seo (seobo576@student.otago.ac.nz)
Alec Holt (alec.holt@otago.ac.nz)
Holger Regenbrecht (holger.regenbrecht@otago.ac.nz)

Version:2 Date: 12 March 2015

Author's response to reviews: see over
Dr Giulia Mangiameli  
Executive Editor  
BMC – Medical Information & Decision Making  

10 March 2015  

Re: Research Article - Web-based Family History Assessment Tools for Colorectal Cancer (CRC) Risk Analysis - Development and Comparison  

Dear Dr Mangiameli,  

Please accept the revised version of our manuscript for publication. We have made the following amendments:  

- email addresses of all authors have been included on the title page.  
- 'Conclusion and Discussion' has been renamed as 'Conclusion'.  
- The manuscript contains required statements regarding ethical approval.  
- The manuscript includes line and page numbers.  
- The manuscript includes a ‘Competing interests’ section.  
- The manuscript contains an ‘Authors’ Contributions’ section.  

Many thanks for consideration of this manuscript.  

Kind Regards,  

A/Prof Michael Schultz